# Net primary production and ecosystem carbon flux of Brazilian tropical savanna ecosystems from eddy covariance and inventory methods

George Louis Vourlitis<sup>1</sup>, Osvaldo Borges Pinto Jr.<sup>2</sup>, Higo José Dalmagro<sup>3</sup>, Paulo Arruda<sup>4</sup>, Francisco de Almeida Lobo<sup>2</sup>, and José de Souza Nogueira<sup>2</sup>

<sup>1</sup>California State University <sup>2</sup>Universidade Federal de Mato Grosso <sup>3</sup>University of Cuiabá <sup>4</sup>Unknown

November 21, 2022

#### Abstract

Estimates of net primary (NPP) and ecosystem production (NEP) are needed for tropical savanna, which is structurally diverse but understudied compared to tropical rainforest. Estimates of NPP and NEP are available from eddy covariance and inventory methods, but both approaches have errors and uncertainties. We used both methods to estimate carbon (C) fluxes for an upland mixed grassland and a seasonally flooded forest to determine the correspondence in C cycling components derived from these methods and assess the contribution of the various C cycling components to the overall NEP. Both techniques provided similar estimates of NPP, NEP, and gross primary production (GPP). Belowground NPP accounted for 49-53% of the total NPP for both ecosystems, followed by aboveground litter (26-27%) and wood (16-17%) production. Increases in water availability increased the potential for C storage, but the mechanism was different in the savanna types with an increase in soil moisture causing higher NPP in the mixed grassland but lower ecosystem respiration ( $R_{eco}$ ) in the Cerrado forest. Compared to other savanna ecosystems, the mixed grassland had a similar rate of  $R_{eco}$  but lower productivity and C use efficiency (CUE = NPP/GPP = 0.28). The Cerrado forest had a high CUE (0.58) and similar C flux rates to other tropical savanna forests and woodlands. While our measurements are spatially and temporally limited, the agreement in C fluxes estimated using inventory and eddy covariance methods suggest that the C cycle estimates for these savanna ecosystems are robust.

#### Hosted file

essoar.10510005.1.docx available at https://authorea.com/users/413202/articles/596607-netprimary-production-and-ecosystem-carbon-flux-of-brazilian-tropical-savanna-ecosystemsfrom-eddy-covariance-and-inventory-methods

## Net primary production and ecosystem carbon flux of Brazilian tropical savanna ecosystems from eddy covariance and inventory methods

George L Vourlitis<sup>1\*</sup>, Osvaldo Borges Pinto Jr.<sup>2</sup>; Higo J. Dalmagro<sup>2</sup>; Paulo Enrique Zanella de Arruda<sup>3</sup>; Francisco de Almeida Lobo<sup>3</sup>; and José de Souza Nogueira<sup>3†</sup>

<sup>1</sup>Department of Biological Sciences, California State University, San Marcos, CA, 92096.

<sup>2</sup>Programa de Pós-Graduação em Ciências Ambiental, Universidade de Cuiabá (UNIC), Cuiabá, Mato Grosso, 78065-900, Brazil.

<sup>3</sup>Programa de Pós-Graduação em Física Ambiental, Universidade Federal de Mato Grosso (UFMT), Cuiabá, Mato Grosso, Brazil.

\*Corresponding author: George L Vourlitis, Department of Biological Sciences, California State University, San Marcos, CA, 92096. Phone: 1-760-750-4119. Email: georgev@csusm.edu.

<sup>†</sup>This paper is dedicated to the memory of our friend and leader Dr. Prof. Jose de Souza Nogueira (Paraná), who lost his battle with cancer on December 12, 2021. The research described here would not be possible without his vision and leadership. His guidance and friendship will be sorely missed, but his legacy lives on with the research groups that he developed and nurtured throughout his life's work.

# Key points

- Estimates of C cycling from inventory and eddy covariance methods were similar for two floristically different tropical savannas of Brazil
- Belowground NPP was 49-53% of the total NPP for both ecosystems, followed by aboveground litter (26-27%) and wood (16-17%) production.
- The grass dominated savanna had a low C use efficiency (CUE = NPP/GPP = 0.28) while the Cerrado forest had a CUE = 0.58.

## Abstract

Estimates of net primary (NPP) and ecosystem production (NEP) are needed for tropical savanna, which is structurally diverse but understudied compared to tropical rainforest. Estimates of NPP and NEP are available from eddy covariance and inventory methods, but both approaches have errors and uncertainties. We used both methods to estimate carbon (C) fluxes for an upland mixed grassland and a seasonally flooded forest to determine the correspondence in C cycling components derived from these methods and assess the contribution of the various C cycling components to the overall NEP. Both techniques provided similar estimates of NPP, NEP, and gross primary production (GPP). Belowground NPP accounted for 49-53% of the total NPP for both ecosystems, followed by above ground litter (26-27%) and wood (16-17%) production. Increases in water availability increased the potential for C storage, but the mechanism was different in the savanna types with an increase in soil moisture causing higher NPP in the mixed grassland but lower ecosystem respiration ( $R_{eco}$ ) in the Cerrado forest. Compared to other savanna ecosystems, the mixed grassland had a similar rate of  $R_{eco}$  but lower productivity and C use efficiency (CUE = NPP/GPP = 0.28). The Cerrado forest had a high CUE (0.58) and similar C flux rates to other tropical savanna forests and woodlands. While our measurements are spatially and temporally limited, the agreement in C fluxes estimated using inventory and eddy covariance methods suggest that the C cycle estimates for these savanna ecosystems are robust.

# Plain language summary

Estimates of net primary and ecosystem production are needed for tropical savannas, which are highly diverse but understudied compared to tropical forests. Several methods have been used to measure primary and ecosystem production; however, these approaches have errors and uncertainties. We used available methods to estimate the productivity of a grass-dominated savanna in the southern Amazon Basin and a seasonally flooded forest in the northern Pantanal to assess the correspondence in the productivity estimated from these methods and evaluate the contribution of above and belowground processes to the overall ecosystem productivity. Productivity estimates derived from the different techniques were similar. Belowground processes accounted for 49-53% of the productivity followed by leaf (26-27%) and wood (16-17%) production. Increases in water availability increased the productivity for both savanna types, but for different reasons. An increase in soil moisture caused an increase productivity in the mixed grassland but a decline in carbon losses from the seasonally flooded forest. Compared to other savanna ecosystems, the mixed grassland had lower productivity while the forest had high productivity. While our measurements are spatially and temporally limited, the broad agreement in the different methods increases our confidence in the productivity estimates for our savanna ecosystems.

*Key words and phrases*: Amazon Basin, Cerrado, biometric measurements, climate change, ecosystems ecology, micrometeorology, Pantanal.

#### 1. Introduction

Savanna makes up 15-37 million km<sup>2</sup> of the global terrestrial surface area (Grace et al. 2006). The wide range associated with this estimate of spatial coverage illustrates the difficulties in defining what constitutes "savanna" due to the high structural complexity of this biome (Hill et al. 2011). For example, Brazilian savanna (known as Cerrado) consists of forests, woodlands, and grass-dominated landcovers, depending on substrate and/or disturbance (fire) regimes, and can be upland or seasonally flooded (Goodland and Pollard 1973; Eiten 1972; Lopes and Cox 1977; Ribeiro and Walter 2008). This structural complexity causes

high plant and animal diversity (Hill et al. 2011) and highly variable estimates of plant and ecosystem production (Grace et al., 2006; Vourlitis and Rocha 2011; Arruda et al. 2016). For example, estimates of net ecosystem  $CO_2$  exchange (NEE) suggest that Cerrado can be a net sink for atmospheric  $CO_2$  (Miranda et al. 1997; Santos et al. 2003), approximately in balance (Rocha et al. 2002), or a net source of  $CO_2$  to the atmosphere (Arruda et al. 2016). The variability in these estimates of NEE is likely due to differences in ecosystem structure, such as tree density, grass cover, leaf area index, climate variability, and hydrology, which influence carbon (C) stocks and C storage capacity (Vourlitis and Rocha 2011; Vourlitis et al. 2013, 2015, 2019).

However, another reason for the high variability in productivity estimates may stem from uncertainties associated with measurement techniques. For example, eddy covariance methods provide continuous measurements of NEE, but localized coverage, differences in sensors, data collection and analysis techniques, and systematic and random errors can increase uncertainty in estimates of ecosystem productivity (Goulden et al. 1996; Miller et al. 2004; Teets et al. 2018). Similarly, rates of net primary production (NPP) and ecosystem production (NEP) can be estimated from field plots using well-known inventory methods (Chen et al. 2003; Grace et al. 2006; Clark et al. 2013), but many components, such as herbivory and below ground production processes, are poorly known and/or rarely measured (Clark et al. 2001a). Furthermore, inventory measurements are time and labor intensive, which severely limits their widespread and/or prolonged use (Teets et al. 2018). Thus, both eddy covariance and inventory measurements of ecosystem production, especially in the floristically diverse tropical savanna, are highly uncertain.

While uncertainties associated with both measurement techniques are increasingly being quantified and reduced (Burba 2013; Chave et al. 2014), simultaneously using both methods can help constrain estimates of NPP and NEP (Miller et al. 2004; Gough et al. 2008; Teets et al. 2018). For example, Miller et al. (2004) used inventory measurements to provide a range of possible net ecosystem productivity estimates for an Amazonian tropical forest to help determine the best frictional velocity (u<sup>\*</sup>) value for reducing errors in nighttime flux loss from inadequate turbulence. Simultaneous flux and tree growth measurements can also shed light on time lags between canopy gas exchange and stand growth (Gough et al. 2008; Teets et al. 2018). Stand inventory data can also be used to understand C cycle dynamics that are not possible from eddy covariance measurements of gas exchange (Clark et al. 2001b).

Here we report simultaneous eddy covariance and stand inventory measurements for two tropical savanna ecosystems in Brazil, a mixed grassland (*campo sujo*) located in the southern Amazon Basin and a seasonally flooded Cerrado forest (*mata de galaria*) located in the northern Pantanal. Data are limited to two years at each site due to challenges associated with maintaining the eddy covariance infrastructure; however, field inventory measurements made over a longer period of time provided data needed for estimating above and belowground NPP and allowing comparisons of C flux estimates derived from eddy covariance and inventory methods. Our objectives were to (1) determine the correspondence between the eddy covariance and inventory measurements, (2) assess the contribution of the various C cycling components to the overall NEP, and (3) to compare the estimates of the C cycling components for our mixed grassland and Cerrado forest to those published in the literature.

## 2. Methods

## 2.1. Site descriptions

Eddy covariance and inventory measurements were conducted in 2011-13 in the Cuiabá Basin and 2015-17 in the Pantanal (Fig. 1). Measurements in the Cuiabá Basin were conducted at the Fazenda Miranda, located 15 km SSE of Cuiabá, Mato Grosso, Brazil (15°43'51"S: 56°04'17"W). The eddy covariance site is situated on a flat plain that is 181 m above sea level and covered primarily by a mixed-grassland (campo sujo) vegetation (Arruda et al. 2016). Soils are acidic (pH = 4.5), stony, shallow (50-100 cm deep), and nutrient-poor red-vellow latosols (Vourlitis et al. 2013). The vegetation is dominated by the tree species Curatella americana and Pouteria ramiflora and exotic  $C_4$  grasses, with a tree density of 533 trees/ha and a grass cover of 65% (Vourlitis et al., 2013; Arruda et al. 2016). Measurements in the Pantanal were made at the Baia das Pedras (16°29'53' 'S; 56°24'46' 'W), which is 130 km SSW from Cuiabá, Mato Grosso, Brazil (Fig. 1). The tower is located 128 m above sea level on a flat floodplain that is seasonally flooded with approximately 50-100 cm of water between December and July (Dalmagro et al. 2019). Soils are classified as Plinthosols (Couto and Oliveira 2011) that have high clay content (50%) but low organic matter (2.0%) and pH (4.8) (Vourlitis et al. 2014). Vegetation is characteristic of a seasonally flooded forest (mata galaria) dominated by Mouriri elliptica, Tabebuia aurea, T. heptaphylla, and Combretum leprosum with a tree density of 1,441 trees/ha and little (5%) understory vegetation (Vourlitis et al. 2014; 2022). Neither site has experienced fire for over 35 years due to active fire suppression.

Both sites have similar climate, with a long-term (30 year) average annual rainfall of 1420 mm and a dry season from May-September when < 5% of the annual rainfall occurs (Nunes da Cunha and Junk 2001; Vourlitis and Rocha 2011). Average annual temperature is 26.5°C, and while seasonal variation is minimal (ca.  $< 2^{\circ}$ C), the wet season is slightly warmer than the dry season (Vourlitis and Rocha 2011).

#### 2.2. Eddy covariance measurements

Tower-based eddy covariance instrumentation, and data collection and analysis methods, were extensively reported for the Cuiabá Basin (Arruda et al. 2016) and Pantanal (Dalmagro et al. 2019). Briefly, both sites used an open-path infrared gas analyzer (LI-7500, LI-COR, Inc., Lincoln, NE, USA) to measure mean and fluctuating quantities of  $CO_2$  and  $H_2O$  vapor density. Measurements of the orthogonal components of wind velocity (ux, uy, uz) in the Pantanal (WindMaster, Gill, Instruments Ltd., Lymington, UK and a Model 81000, R.M. Young

Company, Traverse City, MI) and Cuiabá Basin (C-SAT3, Campbell Scientific, Inc., Logan, UT, USA) were made using three-dimensional sonic anemometers. Eddy covariance sensors were mounted 10 and 20 m above ground in the Cuiabá Basin and Pantanal, respectively, and measurements were made at 10 Hz.

Raw data were processed at 30-min intervals using EddyPro® software (v.6.1.0) (LI-COR Biosciences, Lincoln, Nebraska) in the Pantanal and Alteddy software (version 3.9) in the Cuiabá Basin. Data were corrected for air density fluctuations (Webb et al. 1980), spectral loss (Moncrieff et al., 2004), sonic air temperature humidity (Van Dijk et al., 2004), and inadequate frequency response (Lee et al. 2004). Data were rejected due to sensor or power failure, warning flags generated by the system software, spikes in the sonic and/or gas analyzer data, nonstationary wind speed conditions, and when flux data were outside plausible ranges (Foken et al. 2005). Data collected in the Pantanal were also screened for low windspeed and turbulence conditions ( $u^* < 0.10 \text{ m/s}$ ). System failure and data rejection resulted in the retention of 75-83% of all possible data in the Cuiabá Basin and 70% in the Pantanal. Data gaps were filled by the marginal distribution sampling method as described by Reichstein et al. (2005). Energy balance closure, assessed as the slope of the regression with sensible and latent heat flux (H+Le) as the dependent variable and net radiation minus ground heat flux (Rn-G) as the independent variable was on average 0.71-0.74 for both study sites (Arruda et al., 2016; Dalmagro et al., 2019).

Net ecosystem CO<sub>2</sub> exchange (NEE) was calculated from measurements of eddy flux and  $CO_2$  storage in the Pantanal (Dalmagro et al. 2019); however,  $CO_2$ storage was not measured in the Cuiabá Basin because of the open canopy associated with the mixed grassland (Arruda et al. 2016). NEE was partitioned into gross primary production (GPP) and ecosystem respiration  $(R_{eco})$  for both sites (Table 1) using the "REddyProc" online tool (Reichstein et al. 2005). These methods assume that nighttime NEE =  $R_{eco}$  (i.e., GPP = 0).  $R_{eco}$  is calculated by (1) estimating the temperature sensitivity of  $R_{eco}$  for the entire data set, (2) calculating a seasonal reference respiration rate  $(\rm R_{eco,ref})$  for each week using a 4 day moving window, and (3) calculating  $R_{eco}$  as a function of temperature using the temperature sensitivity and  $R_{eco,ref}$  for all 30-minute periods (Arruda et al., 2016). GPP was then calculated from the measured NEE and estimated  $\mathbf{R}_{\mathrm{eco}}$  values and integrated over an annual basis (Table 1). Autotrophic respiration  $(R_{Aut})$  was estimated as 63% of  $R_{eco}$  based on measurements from other tropical forest (Malhi et al. 2009), and heterotrophic respiration  $(R_{Het}) = R_{eco} - R_A$  (Table 1).

#### 2.3. Meteorological measurements

Meteorological sensors and measurement methods were discussed in detail by Arruda et al. (2016) for the Cuiabá Basin and Dalmagro et al. (2019) for the Pantanal. Air temperature and relative humidity were measured 10 m (Cuiabá Basin) and 20 m (Pantanal) above ground level using a thermohygrometer (HMP45AC, Vaisala Inc., Woburn, MA, USA). These data were used to calculate the vapor pressure deficit (VPD), which was calculated as the difference between the atmospheric saturation vapor pressure minus the actual atmospheric vapor pressure. Solar radiation (Rs) was measured 5 m (Cuiabá Basin) and 20 m (Pantanal) above ground level using a pyranometer (LI200X, LI-COR Biosciences, Inc., Lincoln, NE, USA). Soil moisture was measured using time domain reflectometry probes (n = 2 probes/site) installed 20 cm below the soil surface (CS616-L50, Campbell Scientific, Inc., Logan, UT, USA). Micrometeorological variables were measured every 30 s and stored and processed using a data logger (CR1000, Campbell Scientific, Inc., Logan, UT). Precipitation was measured at both sites; however, the data were not consistently reliable. Thus, precipitation data were obtained for the Cuiabá station (Station No. 83361) from Brazilian Instituto Nacional de Meteorologia (INMET; https://bdmep.inmet.gov.br/; accessed on 1 December, 2021), which is located between both research sites.

#### 2.4. Inventory measurements

Woody C storage measurements ( $P_W$ ; Table 1) were described in detail by Vourlitis et al., (2019, 2022). Briefly, annual measurements of stem density and circumference were made during the dry season (July-August) in 2011-13 for the Cuiabá Basin and 2015-17 for the Pantanal when variations in stem diameter due to trunk water-storage were small (Vourlitis et al. 2022). Tree density was measured at each site using the point-quarter method (Vourlitis et al. 2013) every 5 m along a randomly placed 100 m transect. The four closest trees to each 5 m sample point with a diameter > 3 cm were tagged (n = 80 trees/transect), identified to genus and species, and measured for circumference at breast height (1.3 m aboveground level). All stems were measured for multi-stemmed trees; however, most trees were single stemmed at breast height. Tagged trees that died were noted, and new trees (> 3 cm in diameter) that were closest to the sample point were tagged and identified as above.

Circumference was converted to diameter at breast height (DBH) and aboveground woody biomass was calculated using measured values of DBH and wood density and estimated values of tree height calculated from DBH using established equations for tropical forest trees (Chave et al. 2014; Vourlitis et al. 2019). Annual changes in stem circumference were calculated from annual measurements of circumference at breast height for the Cuiabá Basin and from plastic spring-loaded dendrometer bands in the Pantanal (Vourlitis et al., 2019; 2022). Wood density was measured from tree cores obtained at breast height using an increment borer (n = 146 trees in the Cuiabá Basin and 129 trees in the Pantanal), and wood C concentration was measured from the dried cores using a wet combustion (Walkley-Black) method (Vourlitis et al. 2015). Wood biomass (kg/tree) was converted to C units (kgC/tree) by multiplying aboveground woody biomass by the wood C concentration, and the aboveground woody C pool was calculated by multiplying woody C by the average stand density (kgC/ha). The annual change in the wood C increment ( $P_W$ ) was calculated as the difference in the woody C pool between two consecutive years (Vourlitis et al. 2019).

Above ground litter production (P<sub>AL</sub>; Table 1) was measured at each site between 2014-2018 from 6-8, randomly located 1 m<sup>2</sup> tall litter fall traps constructed from metal frames and 1 mm nylon mesh (Pinto Jr. et al. 2020). Litter fall was collected monthly, dried at 70°C for 72 hours, and separated into leaves, stems, and reproductive structures. Litter was analyzed for C concentration by wet combustion methods described above (Pinto Jr. et al. 2020), and litter dry mass per unit area was converted to C units by multiplying the dry mass by the C concentration. Annual P<sub>AL</sub> was calculated as the sum of the monthly litter produced each year. Measured values of P<sub>AL</sub> for the Cerrado forest (Pinto Jr. et al. 2020) were used to estimate P<sub>AL</sub> for the study period (2015-17). However, P<sub>AL</sub> as not measured in the mixed grassland in 2011-13 and was estimated to be 0.46 MgC m<sup>-2</sup> y<sup>-1</sup> based on the site average measured in 2014-17 (Table 1).

Estimates of residual productivity  $(P_{resid})$  correspond to the sum of herbaceous productivity  $(P_H)$  and the amount of leaf production lost to herbivores and emission of volatile organic compounds (VOC).  $P_H$  was considered as a residual productivity because it was a minor component of the overall productivity for each site (Vallejo 2020).  $P_H$  was measured from each site in 2019 by Vallejo (2020) as the difference between the maximum and minimum herbaceous C pool size, but because these measurements did not correspond to the 2011-13 and 2015-17 study periods,  $\mathrm{P_{H}}$  was estimated for both sites as the ratio of  $\mathrm{P_{W}}$ and  $P_{\rm H}$  calculated in 2019 by Vallejo (2020) (Table 1). The leaf production was estimated from the litterfall data assuming that leaf litter accounted for nearly all (ca. 80%) of the of the annual  $P_{AL}$  (Pinto Jr. et al. 2018; 2020). Measurements of leaf herbivore damage made from two of the most common species found at each study site (*Curatella americana* and *Vochysia divergens*) indicate that herbivores consume on average 25% of the biomass from 28% of the leaves produced at 75% of the biomass from an additional 6% of leaves produced (Dalmolin et al., 2015). VOC emissions were estimated at 2% of annual leaf production (Clark et al. 2001a).

Root biomass was measured in 2013 from both sites in the upper 50 cm soil layer from soil cores collected every 20 m along each randomly selected 100 m transect. Collected roots (> 2 mm fragments) were removed by wet sieving, cleaned of debris, and dried at 72°C. Root density was calculated as the dry root mass per unit core area (g dry weight/ $m^2$ ). Roots were measured for C concentration as described for wood and litter above, and root biomass pool sizes were converted to a C pool sizes by multiplying biomass per unit area by the root C concentration. Because these root C pools were only measured once, annual estimates of root biomass production  $(P_R; Table 1)$  were obtained from surface (0-10 cm) core samples obtained monthly from the Cerrado forest site in 2014-15 (n = 6 cores/site/month; Pinto Jr. et al., 2018). Root biomass per unit area were converted to C pools using the root C concentration measured for surface roots as above, and the  $\mathbf{P}_{\mathbf{R}}$  was calculated as the difference between the maximum and minimum root C pool size over the 2014-15 study period. These estimates of  $P_{\rm R}$  resulted in a root turnover rate of 0.80 y<sup>-1</sup>, which is consistent with that reported for tropical forests worldwide (Gill and Jackson, 2000). Because  $P_R$  was not measured for either site during the study period,  $P_R$  had to be estimated from an existing measured value. We chose  $P_w$ , since it was the most consistently measured C cycle component and there was enough information from our own field measurements and already published studies to provide a justifiable estimate of  $P_R$ . Published results from tropical forests indicated that the ratio of  $P_R$  (coarse and fine) to Pw was between 0.8 and 1.25 (Chambers et al. 2004; Malhi et al. 2009); however, Pinto Jr. et al. (unpublished) and Taques (2020) found a ratio of 1.9 for the seasonally flooded Cerrado forest studied here and Chen et al. (2003) reported a ratio of 4.4 for a eucalypt forest-savanna. We chose an average of 1.7 from these studies (Table 1), which suggests a higher C allocation below ground, as is indicated by our measurements and those of Chen et al. (2003). Belowground litter production (P<sub>BL</sub>) was estimated from  $P_R$  using the mean turnover rate of 0.80 y<sup>-1</sup> calculated for the Cerrado forest described above (Table 1).

#### 2.5. Data analysis

Estimates of total NPP (TNPP), NEE, and GPP were estimated for both eddy covariance and inventory-based methods (Table 1). For eddy covariance methods, annual estimated of NEE and GPP were derived from the eddy covariance measurements as described in section 2.2, while TNPP was calculated as GPP

 $\rm R_{Aut}.$  For inventory methods, TNPP was calculated as the sum of above-ground NPP ( $\rm P_W$  +  $\rm P_{AL}$  +  $\rm P_{resid}$ ) and belowground NPP ( $\rm P_R$  +  $\rm P_{BL}$ ), NEE = TNPP  $\rm R_{Het},$  and GPP = TNPP +  $\rm R_{Aut}$  (Table 1). The annual change in the soil organic C pool (SOC) was estimated as the sum of the above and belowground litter inputs ( $\rm P_{AL}$  +  $\rm P_{BL}$ ) minus heterotrophic respiration ( $\rm R_{Het}).$  Annual changes in above- and belowground C pools (AGC and BGC, respectively) were estimated as the residual of the eddy derived NEE and SOC (NEE

SOC). This residual was partitioned into AGC and BGC pools based on their fraction of the total plant C pool, which is consistent with measurements made in Amazonian tropical forests (Malhi and Grace 2000).

Confidence intervals (95%) were calculated for fluxes based on levels of replication where multiple, independent, randomly samples were available (e.g.,  $P_{AL}$ ,  $P_R$ ). For unreplicated data (e.g.,  $P_W$ ; eddy fluxes), 95% confidence intervals were calculated using bootstrapping randomization techniques as described in Arruda et al. (2016), Dalmagro et al., (2019), and Vourlitis et al. (2019).

#### 3. Results

#### 3.1. Meteorological conditions during the study period

Annual precipitation was on average 1467 mm over the 2011-13 study period (Fig. 2a), which was slightly higher than the long-term average, and 1420 mm over the 2015-16 study period (Fig. 2b), which is identical to the long-term average. However, interannual variations were large, with 2011-12 registering about 200 mm more precipitation than 2012-13 for the Cuiabá Basin and 2015-16 recording nearly 700 mm less rainfall than 2016-17 for the Pantanal. For the Cuiabá Basin, 2011-12 had significantly higher solar radiation than 2011-

13 (Fig. 2c) while 2012-13 had significantly higher soil moisture than 2011-12 (Fig. 2i). These patterns contrast the interannual variations in precipitation and suggest that the wetter year (2011-12) had higher solar radiation and lower soil moisture. However, the reason for the higher precipitation in 2011-12 was due to extremely high rainfall in May 2012 (ca. 264 mm vs. 20 mm in May 2013), which is normally the beginning of the dry season in this region.

For the Pantanal, 2015-16 was significantly warmer (Fig. 2f) and drier, with a significantly higher vapor pressure deficit (VPD; Fig. 2h) and lower soil moisture (Fig. 2j) than 2016-17. These differences can be attributed to the strong El Niño that occurred in 2015-16, which normally cause warmer and drier than average conditions for this region, and the subsequent La Niña that occurred the following year, which normally cause cooler and wetter conditions for the northern Pantanal (Panisset et al., 2018).

3.2. Correspondence between eddy covariance and inventory estimates of C cycling

Estimates of total NPP (TNPP), gross primary production (GPP), and net ecosystem  $CO_2$  exchange (NEE) were similar for inventory and eddy covariance methods with no significant differences (Fig. 3). For both sites, estimates of TNPP were consistently higher for the inventory-based approach than those derived from eddy covariance; however, these differences were not statistically significant (Fig. 3a and b). Inventory methods tended to underestimate the net ecosystem C loss of the mixed grassland in 2012-13 (Fig. 3c) and overestimate the net ecosystem C uptake for the Cerrado forest in 2015-16 (Fig. 3d), but again there were no significant differences between the eddy and inventory estimates. Correspondence between GPP derived from inventory and eddy measurement techniques was qualitatively similar to that observed for TNPP, with slightly higher rates of GPP estimated from inventory methods (Fig 3e and f). However, none of the differences between the GPP estimated from inventory and eddy covariance methods were statistically significant.

3.3. Interannual variations in C cycle

The mixed grassland in the Cuiabá Basin was a consistent net C source to the atmosphere of 2.75-3.30 MgC ha<sup>-1</sup> y<sup>-1</sup> during the 2011-13 study period, while the Cerrado forest in the Pantanal was in balance in 2015-16 and a net sink of approximately 4.75 MgC ha<sup>-1</sup> y<sup>-1</sup> in 2016-17 (Fig. 4a and b; Table 1). Interannual variations in ecosystem respiration ( $R_{eco}$ ) and GPP were minimal for the mixed grassland (Fig. 4c and d; Table 1), but for the Pantanal, the warmer and drier conditions in 2015-16 year caused a significant increase in  $R_{eco}$  than in 2016-17 which was wetter and cooler (Fig. 4f).

In contrast, aboveground NPP (ANPP) was nearly 2-times higher for the mixed grassland in 2012-13 than in 2011-12 while ANPP for the Pantanal was slightly higher (ca. 0.60 MgC ha<sup>-1</sup> y<sup>-1</sup>) in 2015-16 than in 2016-17 (Fig. 5a and b; Table 1). ANPP was 0.74 and 1.21 MgC ha-1 y-1 for the mixed grassland in 2011-12 and 2012-13, respectively and 4.70 and 4.09 MgC ha-1 y-1 for the Cerrado

forest in 2015-16 and 2016-17, respectively (Fig. 5a and b; Table 1). ANPP made up on average 51% of the TNPP for the mixed grassland (Fig. 5e) and 47%for the Cerrado forest (Fig. 5f) depending on year. Belowground net primary production (BNPP) exhibited similar tends as ANPP for the mixed grassland (Fig. 5c), but BNPP for the Cerrado forest was approximately 1.7 MgC ha<sup>-1</sup> y<sup>-1</sup> higher in 2016-17 than in 2015-16 (Fig. 5d). BNPP was between 0.50 and 1.59 MgC ha<sup>-1</sup> y<sup>-1</sup> for the mixed grassland and 4.28 and 5.95 for the MgC ha<sup>-1</sup> y<sup>-1</sup> Cerrado forest (Fig. 5c and d; Table 1), and on average, BNPP accounted for 49 and 53% of the TNPP for the mixed grassland and Cerrado forest, respectively (Fig. 5e and f). In the mixed grassland, root production (PR) and aboveground litter production (PAL) were the dominant terms in TNPP (ca. 27%; Fig. 5e), while in the Cerrado forest,  $\mathbf{P}_{\mathrm{R}}$  accounted for 30% of TNPP followed closely by  $P_{AL}$  (26%; Fig. 5f). After that, below ground litter production ( $P_{BL}$ ) accounted for 22% of the TNPP for the mixed grassland and 24% for the Cerrado forest, wood production  $(P_W)$  16% in the mixed grassland and 17% in the Cerrado forest), and residual production  $(P_{Resid})$  was 8% in the mixed grassland and 3% for the Cerrado forest (Fig. 3e and f).

## 3.4. Mixed grassland and Cerrado forest C cycle

The average GPP for the Cerrado forest from eddy covariance was 16.3 MgC ha<sup>-1</sup> y<sup>-1</sup> for the two year study period, while the TNPP from inventory measurements was on average 10.8 MgC ha<sup>-1</sup> y<sup>-1</sup>, resulting in a C use efficiency (CUE = NPP/GPP) of 0.58 (Fig. 6). The NEE from eddy covariance (2.3 MgC ha<sup>-1</sup> y<sup>-1</sup>) was partitioned into the above, below, and soil organic C (SOC) pools to estimate the instantaneous change in the soil and above and belowground vegetation C pools. First, the change in the SOC pool was calculated as  $[P_{AL}]$ +  $P_{BL}$ ]  $R_{Het}$ , and given these fluxes, the instantaneous change in the SOC pool was estimated to be -0.5 MgC ha<sup>-1</sup> y<sup>-1</sup> (Fig. 6). Next, the remaining NEE was partitioned into the above and belowground C pools by assuming that the change in the pool size is a function of the fractional size of each C pool (Malhi and Grace, 2000). Thus, given the fraction of the aboveground C pool to the total plant C pool [AGC/(AGC+BGC)] = 0.73, the change in the aboveground C pool (AGC) would be 2.1 MgC ha<sup>-1</sup> y<sup>-1</sup> and the change in the belowground C pool (BGC) would be 0.7 MgC ha<sup>-1</sup> y<sup>-1</sup> (Fig. 6). Thus, while the SOC pool is estimated to be declining given the instantaneous rate of R<sub>Het</sub>, the vegetation C pools must be accumulating C given the rate of NEE measured by the eddy covariance tower. These instantaneous rates of vegetation C storage would be larger if the estimate of NEE derived from the inventory methods (ca. 4.3 MgC  $ha^{-1} v^{-1}$ ; Table 2) was used.

In contrast, the GPP for the mixed grassland was 7.5 MgC ha<sup>-1</sup> y<sup>-1</sup> from eddy covariance and the TNPP from inventory measurements was 2.1 MgC ha<sup>-1</sup> y<sup>-1</sup>, resulting in a CUE of 0.28 (Fig. 6). Using the eddy covariance NEE as the reference value, SOC was -2.9 MgC ha<sup>-1</sup> y<sup>-1</sup>, accounting for nearly all of the net ecosystem C loss, and both AGC and BGC were -0.1 MgC ha<sup>-1</sup> y<sup>-1</sup> (Fig. 6).

## 4. Discussion

4.1. Correspondence between the eddy covariance and inventory estimates and other studies

The C flux values calculated from inventory methods agreed well to those estimated from eddy covariance (Fig. 3), and while small, differences between inventory and eddy covariance methods are expected given the inherent measurement errors and uncertainties associated with both methods and time lags between rates of  $CO_2$  exchange and growth (Gough et al. 2008; Teets et al. 2018).

Our estimates of the key C cycle variables for the Cerrado forest were similar to those published in the literature (Table 2). For example, estimates of NEE from other savanna forests and woodlands ranged between 0.1 to 3.8 MgC ha<sup>-1</sup> y<sup>-1</sup>, which agrees with the Cerrado forest estimates derived from both methods (Table 2). However, our mixed grassland was consistently less productive than other comparable grass-dominated tropical savannas during the study period and was losing C (Fig. 6; Table 2). There are undoubtedly many reasons for these differences, but soils associated with our mixed-grassland are shallow, stony, with low water-holding capacity and fertility (Vourlitis et al. 2013), and it is likely that these conditions promote low productivity. Furthermore, fire has been suppressed at this site for decades, and periodic fire can stimulate gross productivity in campo forms of Cerrado (Santos et al. 2003). Interestingly, the only C cycle variable for the mixed-grassland that agreed well with other grass-dominated savannas was  $R_{eco}$ , which was 10.6 MgC ha<sup>-1</sup> y<sup>-1</sup> for our mixedgrassland compared to 9.8 MgC ha<sup>-1</sup> y<sup>-1</sup> for other grass-dominated savanna in Brazil (Table 2).

Estimates of GPP for the Cerrado forest studied here ranged between 16.3 and 18.4 MgC ha<sup>-1</sup> y<sup>-1</sup> based on eddy covariance and inventory estimates, respectively, which is slightly lower than the estimate reported for a eucalypt forest-savanna in Australia (Table 3). However, the estimate of  $R_{eco}$  for our Cerrado forest is well lower than that reported for the same forest (Table 3), which is probably related to the seasonally flooded hydrology of our forest (Dalmagro et al. 2019). However, rates of ANPP, BNPP, and TNPP for our Cerrado forest are well within the range of values reported for other forest- and woodland-savanna (Table 3), which provides confidence in the values estimated here.

4.2. Carbon storage and cycling of Cerrado mixed-grasslands and forests

Carbon use efficiency was on average 0.58 for the Cerrado forest and 0.28 for the mixed grassland, compared to CUE values reported for tropical forests (0.30-0.51; Malhi et al. 1999, 2009; Chambers et al. 2004) and savanna (0.53; Chen et al. 2003). The lower value for the mixed grassland presumably reflects the relatively low growth rates of the C4 grasses and dominant tree species (*Curatella americana*), which has a lower RGR than many other tree species observed the Cuiabá Basin and Pantanal (Vourlitis et al. 2022). In contrast, the high value for the Cerrado forest reflects a greater proportion of GPP that is allocated to growth (Chambers et al. 2004), and dominant tree species within the tower footprint (e.g., *Mouriri elliptica* Mart., *Eugenia florida* DC., *Vochysia divergens*, and *Erythroxylum anguifugum*) have high relative growth rates (Vourlitis et al. 2022). Many of these species are also considered to be pioneer or secondary species, suggesting that the Cerrado forest is recovering from a disturbance that happened in the past.

TNPP was roughly equally partitioned into ANPP and BNPP for both ecosystems studied, which is similar to that reported for other savanna ecosystem worldwide (Table 2). Most of the TNPP was composed of  $P_{\rm R}$ , which reflects the relatively higher belowground C allocation for both arboreal and grassdominated savanna (Chen et al., 2003; Mokany et al. 2006), followed by  $P_{AL}$ ,  $\rm P_{BL}, \, P_{W}, \, and \, P_{Resid}$  accounting for the lowest fraction of TNPP (Fig. 5).  $\rm P_{AL}$ was also an important component of TNPP, exceeding  $P_W$  by 9-11%, and since the majority of  $\mathrm{P}_{\mathrm{AL}}$  is leaf litter (80%; Pinto Jr. et al. 2018; 2020), these results imply an important contribution of leaf production to the total forest NPP. The NPP of tropical forests is typically dominated by leaf production (Clark et al. 2001b; Chambers et al. 2004) reflecting the high leaf area index of these forests.  $P_W$  accounted for 16-17% of the TNPP, but 30-39% of the ANPP for the savanna ecosystems studied here, which is similar to that reported for other tropical forests (Clark et al., 2001b) and savannas (Chen et al. 2003). This C allocation to wood represents a stable, long-term C storage pool (Malhi et al. 2004).

Our data suggest that losses of NPP from herbivory and grazing are a minor component of the TNPP for these systems, but unfortunately, we are unable to reduce the potential uncertainty in our estimates of herbivore pressure on NPP. ANPP lost to herbivores has been shown to be an important avenue of C loss for some of the tree species studied here (Dalmolin et al. 2015), and grazing losses for pastures in the Pantanal-Cerrado transition of Mato Grosso account for around 15% of annual NEE (Dalmagro et al. in prep.). In contrast, Wilsey et al. (2002) found that aboveground biomass production was positively correlated with grazing intensity even though annual NEE was not significantly affected by grazing.

Given the NEE and productivity estimates for each ecosystem type, instantaneous rates of C storage were positive for the above- (AGC) and belowground (BGC) and soil C (SOC) pool for the forest but not for the mixed grassland (Fig. 6). For the forest, inputs of C from above and belowground litter production are apparently not enough to balance C losses from heterotrophic respiration, and as a result, we estimate an instantaneous loss of C from the SOC pool (Fig. 6). To balance the NEE, the largest increase was observed for the AGC pool (+1.6 MgC ha<sup>-1</sup> y<sup>-1</sup>), based on the ratio of above and below ground pool size, followed by the BGC pool. For the grassland, the largest loss was estimated for the SOC pool (-2.9 MgC ha<sup>-1</sup> y<sup>-1</sup>), which almost completely accounted for the net ecosystem loss (Fig. 6). This result, while surprising, may be reasonable in light of the high  $R_{eco}$  (ca. 10.6 MgC ha<sup>-1</sup> y<sup>-1</sup>) estimated for the mixed grassland (Fig. 6), which is about 0.8 MgC ha<sup>-1</sup> y<sup>-1</sup> higher than estimates reported for other grass-dominated topical savannas (Table 2).

#### 4.3. Interannual patterns of C fluxes

Admittedly, the two-year study period associated with each savanna landcover type limits our ability to assess climatic controls on C cycling. However, some trends emerged that are consistent with what is known about how climate variation affects C cycling in these ecosystems. TNPP was higher in 2012-13 for the mixed grassland, due to higher ANPP and BNPP, while NEE was trending higher in 2012-13 because of increases in GPP. While 2012-13 had lower rainfall and solar radiation than 2011-12, soil moisture was significantly higher in 2012-13, which presumably increased water availability. Increases in water availability typically increase leaf photosynthesis (Dalmagro et al. 2014; Dalmolin et al., 2018), NEE (Arruda et al. 2016), tree growth (Vourlitis et al. 2022) and wood C storage (Vourlitis et al. 2019) for Cerrado of the Cuiabá Basin. The interannual variations in NEE and NPP for the Pantanal forest are perhaps more interesting, as net ecosystem  $CO_2$  uptake increased significantly in 2016-17, which was a cooler and wetter year than 2015-17. However, the increase in NEE appeared to be due more to a decline in  $\mathbf{R}_{\mathbf{eco}}$  than an increase in productivity (i.e. GPP or NPP). Growth rates for many of the tree species in the study area have been shown to be sensitive to variations in rainfall dry season length, and temperature (Vourlitis et al., 2022), so it is not clear why ANPP, and thus, TNPP did not increase in 2016-17, which was significantly wetter and cooler. Apparently, the increase in rainfall and soil moisture caused an increase in the duration of the anaerobic period, which reduced  $R_{eco}$  (Dalmagro et al. 2019).

## 4.4. Caveats and limitations

Our estimates of C fluxes are broadly consistent with those reported for similar tropical savanna ecosystems. However, there are several limitations associated with our data that warrant consideration. First, the eddy covariance and inventory measurements only overlapped for two years at each study site, which is probably not enough time to robustly assess temporal trends in the eddy and inventory C fluxes, and especially, climatic controls on these variables (Clark et al. 2013). This is especially true for the inventory methods, which require estimated of litter production for both above and below ground NPP, which were either not measured at either site  $(\mathbf{P}_{\mathrm{BL}})$  or not measured when other inventory measurements were conducted ( $P_{AL}$  in the mixed grassland). The lack of such important data required their estimation from averages measured during other time periods or from other measurements, which have their own uncertainties and errors. Because  $P_{AL}$  was a dominant component for ANPP, the lack of measured data for the mixed grassland presumably increases the uncertainty in the inventory estimates of TNPP and NEE for this landcover type. The same can be said for  $P_R$ , which was estimated empirically from  $P_W$  because comparable data of both sites were lacking. Given the presumed importance of  $P_R$  for both sites, the lack of actual measurements during the study period adds uncertainty to the estimates of BNPP and TNPP.

Estimates of C cycling variables also required assumptions that were highly uncertain. As mentioned above, assumptions of herbivore pressure on NPP were highly uncertain, and estimates of root litter production were based on measured rates of root turnover for the Cerrado forest only. While these estimates of root turnover were consistent with those reported for other topical forests (Gill and Jackson, 2000), it is unclear whether these values are representative of the mixed grassland and/or whether they adequately characterize belowground litter production.

Finally, our scaled estimates of NEE and GPP from the inventory measurements assumed that the tower footprints were dominated by mixed grassland vegetation (Cuiabá Basin) and seasonally flooded Cerrado forest (Pantanal) because detailed analyses of the tower footprints were not conducted. We feel that the lack of a detailed footprint assessment is not likely to significantly increase the uncertainty of our inventory estimates because both Arruda et al. (2016) and Dalmagro et al. (2019) indicated that ca. 80% of the eddy flux originated within 250 m upwind of the flux towers, which consisted of relatively uniform mixed grassland or Cerrado forest vegetation. However, variations in the composition of trees and herbaceous plant species, cover, soil type, and/or C stocks are unknown and may contribute to scaling errors for the inventory data.

## 5. Conclusions

We estimated C fluxes for two tropical savanna ecosystems in the southern Amazon Basin and northern Pantanal using eddy covariance and inventory measurements. Both techniques provided similar estimates of NPP, NEE, and GPP, which increases the level of confidence in the C fluxes estimated for these landcover types. Belowground NPP accounted for approximately half of the total NPP for both ecosystems, followed by aboveground litter and wood production. Carbon fluxes for both savanna types increased with higher water availability but for different reasons; an increase in soil moisture stimulated above and below ground NPP in the mixed grassland but caused a decrease in  $R_{eco}$  in the seasonally flooded Cerrado forest. Compared to other savanna ecosystems, the mixed grassland had low productivity and low C use efficiency (CUE = 0.28) but had similar rates of  $R_{eco}$ . In contrast, the Cerrado forest had a high CUE (0.58), and C flux rates estimated for the Cerrado forest were similar to other savanna forests and woodlands. While our measurements are limited and potentially uncertain, the broad agreement in the C fluxes estimated using inventory and eddy covariance metods, coupled with those reported in the literature, indicate that our estimates of C flux for these savanna ecosystems are robust. Furthermore, the sensitivity of these ecosystems to variations in water availability suggests that warming and drying associated with climate change will likely alter the C productivity and C balance of savanna ecosystems in the southern Amazon Basin and northern Pantanal.

#### Acknowledgements

Research was funded by the Coordination of Improvement of Higher Education

Personnel (CAPES) Science without Borders Program to GV (n. 401908/2012-7), a productivity scholarship funded by the National Council for Scientific and Technological Development and the Ministry of Science and Technology to GV (CNPq n. 304575/2017-8), and the National Science Foundation - Office of International Sciences and Engineering grant to GV (NSF-OISE 0968245). The authors thank the Federal University of Mato Grosso (UFMT - PGFA) and University of Cuiabá (UNIC) for laboratory and office facilities and vehicle access, SESC-Pantanal and Dr. Clovis Miranda and his family for access to the field sites and logistical support, and the Xomanoflux group for the field support. The authors declare no conflicts of interest.

**Data statement:** Data for wood C storage  $(P_W)$  are deposited on Mendeley (https://data.mendeley.com) under the citation: Vourlitis, George (2021), "Spatial and temporal variations in aboveground woody carbon storage for cerrado forests and woodlands of Mato Grosso, Brazil", Mendeley Data, V2, doi: 10.17632/2989vngvys.2. Eddy covariance and meteorological data for the Cerrado forest at Baia das Pedras (Brazil-Northern Pantanal Wetland: BR-Npw) is available through Ameriflux (https://ameriflux.lbl.gov/sites/siteinfo/BR-Npw). Data used in this paper, and selected formulas, are available on Mendeley (https://data.mendeley.com) under the citation: Vourlitis, George (2021), "Net primary production and ecosystem carbon flux of Brazilian tropical savanna ecosystems from eddy covariance and inventory methods", Mendeley Data, V1, doi: 10.17632/kcwwktkd85.1

## Author contributions

Conceptualization: George Vourlitis

Data curation: George Vourlitis, Osvaldo Borges Pinto Jr., Higo J. Dalmagro, Paulo Enrique Zanella de Arruda

Formal Analysis: George Vourlitis, Osvaldo Borges Pinto Jr., Higo J. Dalmagro, Paulo Enrique Zanella de Arruda

Funding acquisition: George Vourlitis

Investigation: George Vourlitis, Osvaldo Borges Pinto Jr., Higo J. Dalmagro, Paulo Enrique Zanella de Arruda, Francisco de Almeida Lobo; and José de Souza Nogueira

Methodology: George Vourlitis, Osvaldo Borges Pinto Jr., Higo J. Dalmagro, Paulo Enrique Zanella de Arruda

Project administration: George Vourlitis, José de Souza Nogueira

Resources: Osvaldo Borges Pinto Jr., José de Souza Nogueira

Visualization: George Vourlitis,

Writing – original draft: George Vourlitis, Osvaldo Borges Pinto Jr., Higo J. Dalmagro, Paulo Enrique Zanella de Arruda

Writing – review & editing: Osvaldo Borges Pinto Jr., Higo J. Dalmagro, Francisco de Almeida Lobo

#### Literature cited

Arruda, P. H. Z., Vourlitis, G. L., Pinto, Jr., O. B., Lobo, F. A., Santanna, F. N., & Nogueira, J. S. (2016) Large net CO<sub>2</sub> loss from a grass-dominated tropical savanna in south-central Brazil in response to seasonal and interannual drought. *Journal of Geophysical Research-Biogeosciences* 121: 2110–2124. doi:10.1002/2016JG003404

Atjay, G. L. (1979) The Global Carbon Cycle. In SCOPE Report 13 (129-181). John Wiley and Sons, Inc., New York, NY.

Burba, G. (2013). Eddy Covariance Method for Scientific, Industrial, Agricultural, and Regulatory Applications. LI-COR Biosciences, Lincoln, NE.

Chambers J. Q., et al. (2004), Respiration from a tropical forest ecosystem: Partitioning of sources and low carbon use efficiency, *Ecol. Appl.*, 14, S72–S88.

Chave, J., Rejou-Mechain, M., Burquez, A., Chidumayo, E., Colgan, M. S., Deltiti, W. B. C., et al. (2014). Improved allometric models to estimate the aboveground biomass of tropical trees. *Global Change Biology*, 20(10), 3177–3190. https://doi.org/10.1111/gcb.12629

Chen, X., Hutley, L. B., & Eamus, D. (2003). Carbon balance of a tropical savanna of northern Australia. Oecologia, 137(3), 405-416. https://doi.org/10.1007/s00442-003-1358-5

Clark, D. A., Brown, S. Kicklighter, D.W., Chambers, J. Q., Thomlinson, J. R., & Ni, J. (2001a). Measuring net primary production in forests: Concepts and field methods. *Ecological Applications*, 11, 356–370, Article Stable URL: http://www.jstor.org/stable/3060894.

Clark, D. A., Brown, S. Kicklighter, D.W., Chambers, J. Q., Thomlinson, J. R., Ni, J., & Holland, E. A. (2001b). Net primary production in tropical forests: An evaluation and synthesis of existing field data. *Ecological Applications*, 11, 371–384, Article Stable URL: http://www.jstor.org/stable/3060895.

Clark, D. A., Clark, D. B., & Oberbauer, S. F. (2013). Field-quantified responses of tropical rainforest aboveground productivity to increasing CO2 and climatic stress, 1997–2009. *Journal of Geophysical Research: Biogeosciences*, 118, 783–794. https://doi.org/10.1002/jgrg.20067

Couto, E. G., & Oliveira, V. A. (2011). The soil diversity of the Pantanal. In W. J. Junk, C. J. da Silva, C. Nunes da Cunha, & K. M. Wantzen (Eds.), *The Pantanal: Ecology, biodiversity, and sustainable management of a large neotropical seasonal wetland*, (pp. 71–102). Sofia-Moscow: Pensoft Publishers.

Dalmagro, H. J., Lobo, F. A., Vourlitis, G. L., Dalmolin, Â. C., Antunes, M. Z. Jr., Ortíz, C. E. R., & Nogueira, J. S. (2014). The physiological light response

of two tree species across a hydrologic gradient in Brazilian savanna (Cerrado). *Photosynthetica*, 52, 22–35.

Dalmagro, H. J., de Arruda, P. H. Z., Vourlitis, G. L., Lathuillière, M. J., Nogueira, J. S., Couto, E. G., & Johnson, M. S. (2019). Radiative forcing of methane fluxes offsets net carbon dioxide uptake for a tropical flooded forest. *Global Change Biology*, 25(6), 1967–1981. https://doi.org/10.1111/gcb.14615

Dalmolin, Â. C., Lobo, F. A., Vourlitis, G., Silva, P. R., Dalmagro, H. J., Antunes Jr., M. Z., & Ortíz, C. E. R. (2015) Is the dry season an important driver of phenology and growth for two Brazilian savanna tree species with contrasting leaf habits?. *Plant Ecology* 216: 407–417. DOI: 10.1007/s11258-014-0445-5

Dalmolin, A. C., Lobo, F. A., Vourlitis, G. L., Dalmagro, H. J., Antunes Junior, M. Z., & Rodriguez-Ortiz, C. E. (2018). Physiological adjustments of an invasive tree species to extreme hydrological events in a tropical seasonal wetland. *Trees*, 32(5), 1365–1375. https://doi.org/10.1007/s00468-018-1718-8

Eamus, D., Hutley, L. B., & O'Grady, A. P. (2001) Daily and seasonal patterns of carbon and water fluxes above a north Australian savanna. *Tree Physiol* 21:977-988.

Eiten, G. (1972). The cerrado vegetation of Brazil. *Botanical Review*, 38, 201–341.

Foken, T., M. Göockede, M. Mauder, L. Mahrt, B. Amiro, and W. Munger (2005), Post-field data quality control. In *Handbook of Micrometeorology*, (Chapter 9, 181–208). Springer, Netherlands, doi:10.1007/1-4020-2265-4\_9.

Gill, R.A. & Jackson, R. B. (2000) Global patterns of root turnover for terrestrial ecosystems. *New Phytol.*, 147, 13-31.

Goodland, R. J., & Pollard, R. (1973). The Brazilian cerrado vegetation: A fertility gradient. *Ecology*, 61, 219–224.

Gough, C.M., Vogel, C.S., Schmid, H.P., Su, H.-B., & Curtis, P.S., 2008. Multiyear convergence of biometric and meteorological estimates of forest carbon storage. Agric. For. Meteorol. 148, 158–170. http://dx.doi.org/10.1016/j.agr formet.2007.08.004.

Goulden, M. L., Munger, J. W., Fan, S. M., Daube, B. C., & Wofsy, S. C. (1996) Measurements of carbon sequestration by long-term eddy covariance: methods and a critical evaluation of accuracy. Global Change Biology 2:169–182.

Grace, J., Jose, J. S., Meir, P., Miranda, H. S., & Montes, R. A. (2006). Productivity and carbon fluxes of tropical savannas. *Journal of Biogeography*, 33, 387–400.

Hill, M. J., Román, M. O. & Schaaf, C. B. (2011) Biogeography and dynamics of global tropical and subtropical savannas: A spatiotemporal view. In *Ecosystem* 

function in global savannas: Measurement and modeling at landscape to global scales, (Chapter 1, 3–37). Boca Raton, FL, USA: CRC, Inc.

House, J. I., & Hall, D. O. (2001) Productivity of tropical grasslands and savannas. In *Terrestrial Global Productivity*. (Chapter 16, 363-400). Academic Press, San Diego.

Lee, X., Massman, W., & Law, B. (2004) Handbook of micrometeorology: A guide for surface flux measurement and analysis. Dordrecht, the Netherlands: Kluwer Academic.

Lopes, A. S., & Cox, F. R. (1977). Cerrado vegetation in Brazil: An edaphic gradient. *Agronomy Journal*, 69, 828–831.

Malhi, Y., Baldocchi, D. D., & Jarvis, P. G. (1999) The carbon balance of tropical, temperate and boreal forests. *Plant Cell Environ* 22:15-740.

Malhi, Y., & Grace, J. (2000) Tropical forests and atmospheric carbon dioxide. Trends in Ecology and Evolution 15:332–337.

Malhi, Y. Saatchi, S., Girardin, C., & Aragão, L. E. O. C. (2009) The production, storage, and flow of carbon in Amazonian forests. In *Amazonia and Global Change* (Chapter 9, 355-372). Geophysical Monograph Series 186, American Geophysical Union, Washington, D.C. 10.1029/2008GM000779.

Miller, S. D., M. L. Goulden, M. C. Menton, et al. 2004. Biometric and micrometeorological measurements of tropical forest carbon balance. *Ecological Applications* 14, S114–S126.

Miranda, A. C., Miranda, H. S., Lloyd, J., Grace, J., Francey, R. J., McIntyre, J. A., et al. (1997). Fluxes of carbon, water and energy over Brazilian Cerrado: An analysis using eddy covariance and stable isotopes. *Plant, Cell & Environment*, 20(3), 315–328. https://doi.org/10.1046/j.1365-3040.1997.d01-80.x

Mokany, K., Raison, R.J., & Prokushkin, A.D. (2006) Critical analysis of root:shoot ratios in terrestrial biomes. Global Change Biology, 12, 84–96, doi: 10.1111/j.1365-2486.2005.001043.x.

Moncrieff, J., Clement, R., Finnigan, J., & Meyers, T. (2004) Averaging, detrending, and filtering of eddy covariance time series. In *Handbook of Micrometeorology* (Chapter 1, 7–31). Dordrecht, the Netherlands: Kluwer Academic.

Nunes da Cunha, C., & Junk, W. J. (2001). Distribution of woody plant communities along the flood gradient in the Pantanal of Pocone, Mato Grosso, Brazil. *International Journal of Ecology and Environmental Sciences*, 27, 63–70.

Olson L.S., Watts, J. A., & Allison, L. J. (1983) Carbon in live vegetation of major world ecosystems. Report ORNL-5862. Oak Ridge National Laboratory, Oak Ridge, TN.

Panisset, J. S., Libonati, R., Gouveia, C. M. P., Machado-Silva, F., França, D. A., França, J. R. A., & Peresa, L. F. (2018). Contrasting patterns of the extreme

drought episodes of 2005, 2010 and 2015 in the Amazon Basin. International Journal of Climatology, 38, 1096–1104.

Pinto, Jr., O. B., Marques, A. C. A., & Vourlitis, G. L. (2020) Aboveground Carbon Storage and Cycling of Flooded and Upland Forests of the Brazilian Pantanal. *Forests*, 11, 665. doi:10.3390/f11060665.

Pinto Jr., O. B., Vourlitis, G. L., Carneiro, E. M. S., Dias, M. F., Hentz, C., & Nogueira, J. S. (2018) Interactions between vegetation, hydrology, and litter inputs on decomposition and soil  $CO_2$  efflux of tropical forests in the Brazilian Pantanal. *Forests* 2018, 9, 281; doi:10.3390/f9050281.

Ribeiro, J. F., & Walter, B. M. T. (2008). As principais fitofisionomias do bioma Cerrado. Pages 151–212 in Sano SM. In S. Pedrosa de Almeida, & J. F. Ribeiro (Eds.), Cerrado Ecologia e Flora, Vol. 1. Emprapa Informacao Technologica, Ministerio da Agricultura, Pecuaria e Abastecimento, Brasilia, (Chapter 6, 153– 212). Districto Federal, Brasil: Embrapa.

Reichstein, M., et al. (2005), On the separation of net ecosystem exchange into assimilation and ecosystem respiration: Review and improved algorithm, *Global Change Biol.*, 11, 1424–1439.

Rocha, H. R., H. C. Freitas, R. Rosolem, et al. (2002) Measurements of  $CO_2$  exchange over a woodland savanna (Cerrado Sensu stricto) in southeast Brazil. *Biota Neotropica* 2, 1–11.

Santos, A. J. B., Silva, G. T. D. A., Miranda, H. S., Miranda, A. C., & Lloyd, J. (2003). Effects of fire on surface carbon, energy and water vapour fluxes over campo sujo savanna in central Brazil. *Functional Ecology*, 17, 711–719.

Scholes, R. J., & Hall, D. O. (1996) The carbon budget of tropical savannas, woodlands and grasslands. In Global change: Effects on coniferous forests and grassland (Chapter 4, 69-100). Wiley, New York.

Taques, A. C. F. S (2020) Classificação da biomassa radicular e monitoramento da serapilheira em áreas do Pantanal e cerrado mato-grossense períodos seco e chuvoso. MS Thesis, Programa en Ciencias Ambientais, Universidade de Cuiabá. pp. 70.

Teets, A., Fraver, S., Hollinger, D. Y., Weiskittel, A. R., Seymour, R. S., & Richardson, A. D. (2018) Linking annual tree growth with eddy-flux measures of net ecosystem productivity across twenty years of observation in a mixed conifer forest. Agricultural and Forest Meteorology, 249, 479–487.

Vallejo, I. (2020) Estimates of, and Influences on, Aboveground Biomass Production of Tropical Savanna (Cerrado) in the Cuiabá Basin and Pantanal of Mato Grosso, Brazil. MS Thesis, California State University, San Marcos, CA, USA. pp. 37.

Van Dijk, A., Moene, A., & De Bruin, H. (2004). The principles of surface flux physics: Theory, practice and description of the ECPACK library. *Internal* 

# Report, 1, 99.

Vourlitis, G. L., & Rocha, H. R. (2011). Flux dynamics in the cerrado and cerrado-forest transition of Brazil. In *Ecosystem function in global savannas: Measurement and modeling at landscape to global scales*, (Chapter 5, 97–116). Boca Raton, FL, USA: CRC, Inc.

Vourlitis, G. L., Lobo, F. A., Lawrence, S., Lucena, I. C., Borges, O. P. Jr., Dalmagro, H. J., et al. (2013). Variations in stand structure and diversity along a soil fertility gradient in a Brazilian savanna (Cerrado) in southern Mato Grosso. Soil Science Society of America Journal, 77, 1370–1379.

Vourlitis, G. L., Lobo, F. A., Lawrence, S., Holt, K., Zappia, A., Pinto, O. B. Jr., & Nogueira, J. S. (2014). Nutrient resorption in tropical savanna forests and woodlands of central Brazil. *Plant Ecology*, 215(9), 963–975. https://doi.org/10.1007/s11258-014-0348-5

Vourlitis, G. L., Lobo, F. A., Pinto, O. B. Jr., Dalmagro, H. J., Arruda, P. H. Z., & Nogueira, J. S. (2015). Variations in aboveground vegetation structure along a nutrient availability gradient in the Brazilian Pantanal. *Plant and Soil*, 389(1-2), 307–321. https://doi.org/10.1007/s11104-014-2364-6

Vourlitis, G. L., Zappia, A., Pinto Jr., O. B., Arruda, P. H. Z., Santanna, F. B., Dalmagro, H. J., Lobo, F. A., & Nogueira, J. S. (2019) Spatial and temporal variations in aboveground woody carbon storage for cerrado forests and wood-lands of Mato Grosso, Brazil. *Journal of Geophysical Research: Biogeosciences*, 124. https://doi.org/10.1029/2019JG005201.

Vourlitis, G. L., Pinto Jr., O. B., Dalmagro, H. J., Arruda, P. E. Z., Lobo, F. A., Nogueira, J. S. (2022) Tree growth responses to climate variation in upland and seasonally flooded forests and woodlands of the Cerrado-Pantanal transition of Brazil. *Forest Ecology and Management*, 505 (2022), 119917. https://doi.org/10.1016/j.foreco.2021.119917.

Wilsey, B. J., Parent, G., Roulet, N. T., Moore, T. R., & Potvin, C. (2002) Tropical pasture carbon cycling: relationships between C source/sink strength, above-ground biomass and grazing. Ecology Letters, 5, 367–376

Webb, E. K., Pearman, G.I., & Leuning, R. (1980), Correction of flux measurements for density effects due to heat and water vapour transfer. *Q. J. R. Meteorol. Soc.*, 106(447), 85–100, doi:10.1002/qj.49710644707.

#### **Figure legends**

**Fig. 1.** Location of the study sites in the Cuiabá Basin (Fazenda Miranda) and the northern Pantanal (Baia das Pedras) (main image) and the location of the study region with respect to Brazil and the South America (small image in the lower right-hand corner). The light-green shaped portion of the main image displays the boundary of the Pantanal. Modified from Justica Ambiental (www.justicaambiental.org.br).

Fig. 2. Meteorological conditions for the Cuiabá Basin (left-panels) and the Pantanal (right-panels). Data are for total annual rainfall (a and b) and mean  $(\pm 95\%$  confidence interval) solar radiation (c and d), air temperature (e and f), vapor pressure deficit (g and h), and soil moisture (I and j). Precipitation data were obtained from INMET for Cuiabá station (Station No. 83361). All other data are from the tower sites. Confidence intervals were obtained by bootstrapping randomization techniques calculated over 1000 iterations. Asterix indicate significant (p<0.05) differences between years.

Fig. 3. Mean ( $\pm 95\%$  confidence interval) (a and b) total net primary production (TNPP), (c and d) net ecosystem CO<sub>2</sub> exchange (NEE), and (e and f) gross primary production (GPP) derived from eddy covariance and increment measurement techniques for the mixed grassland in the Cuiabá Basin (left-panels) and the Cerrado forest in the Pantanal (right-panels).

Fig. 4. Mean ( $\pm$  95% confidence interval) net ecosystem CO<sub>2</sub> exchange (NEE: a and b), gross primary production (GPP: c and d), and ecosystems respiration (e and f) for the Cuiabá Basin (left-panels) and the Pantanal (right-panels). Positive values indicate net CO<sub>2</sub> gain by the ecosystem. Confidence intervals were obtained by bootstrapping randomization techniques calculated over 1000 iterations. Asterix indicate significant (p<0.05) differences between years.

Fig. 5. Net primary production (NPP) components for aboveground NPP (ANPP: a and b), belowground NPP (BNPP: c and d), and the of wood ( $P_W$ ), aboveground litter ( $P_{al}$ ), residual ( $P_{resid}$ ), root ( $P_r$ ), and belowground litter (Pbl) production to TNPP for mixed-grasslands of the Cuiabá Basin (left-panels) and Cerrado forests of the Pantanal (right-panels).  $P_W =$  wood production;  $P_{al} =$  aboveground litter production;  $P_{resid} =$  residual production (herbaceous production + production losses to herbivory and volatile organic C emission);  $P_r =$  root production;  $P_{bl} =$  belowground litter production.

Fig. 6. Conceptual carbon balance for the Cerrado forest in the Pantanal (left-panel) and the mixed grassland in the Cuiabá Basin (right-panel) based on a combination of stand increment and eddy covariance measurements for 2011-13 (Cuiabá Basin) and 2015-17 (Pantanal). Fluxes (MgC ha<sup>-1</sup> y<sup>-1</sup>) are displayed as arrows, with gains to the ecosystem as green arrows and losses as red arrows. Pools (boxes: MgC/ha) are based on data published by Pinto Jr. et al. (2020), Vourlitis et al. (2013, 2015, and 2019), and from unpublished data. Data are averages of those displayed in Table 1. The change in soil organic C (SOC) is estimated as the sum of the above and belowground litter inputs minus heterotrophic respiration. The change in above- and belowground C pools (AGC and BGC, respectively) are estimated as NEE - SOC and are partitioned into AGC and BGC pools based on their fraction of the total plant C pool. Positive values of net ecosystem CO<sub>2</sub> exchange (NEE) depict net uptake by the ecosystem.

Table 1. Data sources and rationale for calculating net ecosystem exchange (NEE), gross primary production (GPP), ecosystem respiration  $(R_{eco})$ ,wood

production ( $P_W$ ), above ( $P_{AL}$ ) and below ground litter ( $P_{BL}$ ) production, herbaceous production ( $P_H$ ), carbon losses from herbivory and volatile organic carbon (VOC) emissions, above ground net primary production (ANPP), root production ( $P_R$ ), total net primary production (TNPP) derived from inventory (I) or eddy covariance (E) methods, autotrophic respiration ( $R_{Aut}$ ), and the estimated NEE derived from inventory methods (NEE<sub>I</sub>). All values are in MgC ha<sup>-1</sup> y<sup>-1</sup>. NEE data with a negative sign indicates net loss from the ecosystem to the atmosphere. See Methods for a detailed explanation of calculation procedures and rationale.

| Variable                                   |   | Cuiabá<br>Basin | Pantanal |         |         |
|--|---|-----------------|----------|---------|---------|
| (MgC<br>ha <sup>-1</sup> y <sup>-1</sup> ) | Data<br>source or<br>calcula-<br>tion                                 | 2011-12         | 2012-13  | 2015-16 | 2016-17 |
| $\text{NEE}_{\text{E}}$                    | CB:<br>Arruda et<br>al. (2016)<br>PAN:<br>Dalmagro<br>et al<br>(2019) |                 |          |         |         |
| $\operatorname{GPP}_{\mathrm{E}}$          | CB:<br>Arruda et<br>al. (2016)<br>PAN:<br>Dalmagro<br>et al<br>(2019) |                 |          |         |         |
| R <sub>eco</sub>                           | CB:<br>Arruda et<br>al. (2016)<br>PAN:<br>Dalmagro<br>et al<br>(2019) |                 |          |         |         |
| $\mathbf{R}_{\mathrm{Aut}}$                | % of $R_{eco}$<br>(Mahli et<br>al. (2009)                             |                 |          |         |         |
| $\rm R_{Het}$                              | Reco -<br>R <sub>Aut</sub>  |                 |          |         |         |
| $\mathbf{P}_{\mathbf{W}}$                  | Vourlitis et<br>al. (2019a<br>and b)                                  |                 |          |         |         |

| Variable                      |                          | Cuiabá<br>Basin | Pantanal |  |
|-------------------------------|--------------------------|-----------------|----------|--|
| P <sub>AL</sub>               | CB:                      |                 |          |  |
|                               | average                  |                 |          |  |
|                               | from                     |                 |          |  |
|                               | 2014 - 2017              |                 |          |  |
|                               | PAN:                     |                 |          |  |
|                               | Pinto Jr.                |                 |          |  |
|                               | et al.                   |                 |          |  |
|                               | (2020)                   |                 |          |  |
| $P_{\rm H}$                   | Vallejo                  |                 |          |  |
|                               | (2020)                   |                 |          |  |
| Herbivory                     | Assumes                  |                 |          |  |
|                               | 28% of                   |                 |          |  |
|                               | leaves =                 |                 |          |  |
|                               | 25% loss                 |                 |          |  |
|                               | and $6\%$ of             |                 |          |  |
|                               | leaves =                 |                 |          |  |
|                               | 75% loss                 |                 |          |  |
|                               | (Dalmolin                |                 |          |  |
|                               | et al.                   |                 |          |  |
|                               | 2016)                    |                 |          |  |
| VOC                           | % of leaf                |                 |          |  |
|                               | production               |                 |          |  |
|                               | (Clark et                |                 |          |  |
|                               | al. 2001a)               |                 |          |  |
| $\mathbf{P}_{\mathrm{Resid}}$ | $P_H +$                  |                 |          |  |
|                               | Herbivory                |                 |          |  |
|                               | + VOC                    |                 |          |  |
| ANPP                          | $P_W + P_{AL}$           |                 |          |  |
|                               | $+ P_{\text{Resid}}$     |                 |          |  |
| $P_R$                         | $P_R = 1.7$              |                 |          |  |
|                               | $\times P_W$             |                 |          |  |
|                               | (Taques                  |                 |          |  |
|                               | 2020)                    |                 |          |  |
| $\mathbf{P}_{\mathrm{BL}}$    | Assumes a                |                 |          |  |
|                               | $P_R$                    |                 |          |  |
|                               | turnover =               |                 |          |  |
|                               | $0.8 \text{ y}^{-1}$     |                 |          |  |
|                               | (Gill and                |                 |          |  |
|                               | Jackson                  |                 |          |  |
|                               | 2000);                   |                 |          |  |
|                               | (Taques                  |                 |          |  |
|                               | 2020)                    |                 |          |  |
| BNPP                          | $P_{\rm B} = P_{\rm BL}$ |                 |          |  |

| Variable   |   | Cuiabá<br>Basin | Pantanal |
|--|---|-----------------|----------|
| TNPPI  | ANPP +<br>BNPP                            |                 |          |
| $\frac{\text{TNPP}_{\text{E}}}{\text{NEE}_{\text{I}}}$ | GPP R <sub>Aut</sub><br>TNPP<br>B         |                 |          |
| GPPI   | ${ m R}_{ m Het}  m TNPP +  m R_{ m Aut}$ |                 |          |

**Table 2.** Summary of published estimates of net ecosystem CO2 exchange (NEE), gross primary production (GPP), ecosystem respiration ( $R_{eco}$ ), aboveground net primary production (ANPP), belowground NPP (BNPP), and total NPP (TNPP) for various savanna ecosystems. Estimates of NEE < 0 indicate net C loss from the ecosystem to the atmosphere. E = eddy covariance; I = inventory or harvest methods.

| Variable                    | MgC ha <sup>-1</sup> y <sup>-1</sup> | Ecosystem type          | Method       | Location  | Reference                    |
|-----------------------------|--------------------------------------|-------------------------|--------------|-----------|------------------------------|
| NEE                         | 2.3                                  | Cerrado forest          | Е            | Brazil    | This study                   |
|                             | 4.3                                  | Cerrado forest          | Ι            | Brazil    | This study                   |
|                             | -3                                   | Mixed grassland         | Е            | Brazil    | This study                   |
|                             | -1.9                                 | Mixed grassland         | Ι            | Brazil    | This study                   |
|                             | 2.8                                  | Mesic savanna forest    | $\mathbf{E}$ | Australia | Eamus et al. $(2001)$        |
|                             | 3.8                                  | Eucalypt forest-savanna | Ι            | Australia | Chen et al. $(2003)$         |
|                             | 2.9                                  | Mixed grassland         | Е            | Brazil    | Santos et al. $(2003)$       |
|                             | 0.1                                  | Woodland savanna        | Е            | Brazil    | Rocha et al. $(2002)$        |
| GPP                         | 16.3                                 | Cerrado forest          | Е            | Brazil    | This study                   |
|                             | 18.4                                 | Cerrado forest          | Ι            | Brazil    | This study                   |
|                             | 7.5                                  | Mixed grassland         | Е            | Brazil    | This study                   |
|                             | 8.7                                  | Mixed grassland         | Ι            | Brazil    | This study                   |
|                             | 20.2                                 | Eucalypt forest-savanna | Ι            | Australia | Chen et al. $(2003)$         |
|                             | 12.7                                 | Mixed grassland         | E            | Brazil    | Santos et al. $(2003)$       |
| $\mathbf{R}_{\mathbf{eco}}$ | 14.1                                 | Cerrado forest          | E            | Brazil    | This study                   |
|                             | 10.6                                 | Mixed grassland         | E            | Brazil    | This study                   |
|                             | 17                                   | Eucalypt forest-savanna | Ι            | Australia | Chen et al. $(2003)$         |
|                             | 9.8                                  | Mixed grassland         | E            | Brazil    | Santos et al. $(2003)$       |
| ANPP                        | 4.4                                  | Cerrado forest          | Ι            | Brazil    | This study                   |
|                             | 1.0                                  | Mixed grassland         | Ι            | Brazil    | This study                   |
|                             | 3.0                                  | Eucalypt forest-savanna | Ι            | Australia | Chen et al. $(2003)$         |
|                             | $6.6 \pm 2.7^{\dagger \ddagger}$     | Savanna                 | Ι            | Various   | House and Hall $(2001)$ (n = |
| BNPP                        | 5.1                                  | Cerrado forest          | Ι            | Brazil    | This study                   |
|                             | 1.0                                  | Mixed grassland         | Ι            | Brazil    | This study                   |
|                             | 8.0                                  | Eucalypt forest-savanna | Ι            | Australia | Chen et al. $(2003)$         |
|                             | $4.6 \pm 2.4^{\dagger \ddagger}$     | Savanna                 | Ι            | Various   | House and Hall $(2001)$ (n = |

| Variable | MgC ha <sup>-1</sup> y <sup>-1</sup> | Ecosystem type          | Method | Location  | Reference                     |
|----------|--------------------------------------|-------------------------|--------|-----------|-------------------------------|
| TNPP     | 7.5                                  | Cerrado forest          | Е      | Brazil    | This study                    |
|          | 9.5                                  | Cerrado forest          | Ι      | Brazil    | This study                    |
|          | 0.9                                  | Mixed grassland         | Е      | Brazil    | This study                    |
|          | 2.0                                  | Mixed grassland         | Ι      | Brazil    | This study                    |
|          | $10.5^{\ddagger}$                    | Forest/shrub woodland   | Ι      | Various   | Atjay et al. $(1979)$         |
|          | $11.5^{\ddagger}$                    | Savanna                 | Ι      | Various   | Atjay et al. $(1979)$         |
|          | $6.4^{\ddagger}$                     | Dry forest              | Ι      | Various   | Olson et al. $(1983)$         |
|          | $5.5^{\ddagger}$                     | Savanna                 | Ι      | Various   | Olson et al. $(1983)$         |
|          | $6.3^{\ddagger}$                     | Woodlands               | Ι      | Africa    | Scholes and Hall (1996)       |
|          | $7.1^{\ddagger}$                     | Savanna                 | Ι      | Africa    | Scholes and Hall (1996)       |
|          | $11.2 + 4.4^{\dagger \ddagger}$      | Savanna                 | Ι      | Various   | House and Hall $(2001)$ (n =  |
|          | 11.0                                 | Eucalypt forest-savanna | Ι      | Australia | Chen et al. $(2003)$          |
|          | $10.2 + 5.9^{\dagger}$               | Forest/woodland         | Ι      | Various   | Grace et al. $(2006)$ (n = 10 |
|          | $5.9 \pm 3.6^{\dagger}$              | Savanna                 | Ι      | Various   | Grace et al. $(2006)$ (n = 12 |

 $^{\dagger}\mathrm{Mean}$   $\pm$  standard deviation

 $^{\ddagger}\mathrm{Data}$  reported as dry mass. Converted to C units assuming the dry mass is 50% C.



**Fig. 1.** Location of the study sites in the Cuiabá Basin (Fazenda Miranda) and the northern Pantanal (Baia das Pedras) (main image) and the location of the study region with respect to Brazil and the South America (small image in the lower right-hand corner). The light-green shaped portion of the main image displays the boundary of the Pantanal. Modified from Justiça Ambiental (www.justicaambiental.org.br)



Fig. 2. Meteorological conditions for the Cuiabá Basin (left-panels) and the Pantanal (right-panels). Data are for total annual rainfall (a and b) and mean  $(\pm 95\%$  confidence interval) solar radiation (c and d), air temperature (e and f), vapor pressure deficit (g and h), and soil moisture (I and j). Precipitation data were obtained from INMET for Cuiabá station (Station No. 83361). All other data are from the tower sites. Confidence intervals were obtained by bootstrapping randomization techniques calculated over 1000 iterations. Asterix indicate significant (p<0.05) differences between years.



Fig. 3. Mean ( $\pm 95\%$  confidence interval) (a and b) total net primary production (TNPP), (c and d) net ecosystem CO<sub>2</sub> exchange (NEE), and (e and f) gross primary production (GPP) derived from eddy covariance and increment measurement techniques for the mixed grassland in the Cuiabá Basin (left-panels) and the Cerrado forest in the Pantanal (right-panels).



Fig. 4. Mean ( $\pm$  95% confidence interval) net ecosystem CO<sub>2</sub> exchange (NEE: a and b), gross primary production (GPP: c and d), and ecosystems respiration (e and f) for the Cuiabá Basin (left-panels) and the Pantanal (right-panels). Positive values indicate net CO<sub>2</sub> gain by the ecosystem. Confidence intervals were obtained by bootstrapping randomization techniques calculated over 1000 iterations. Asterix indicate significant (p<0.05) differences between years.



Fig. 5. Net primary production (NPP) components for aboveground NPP (ANPP: a and b), belowground NPP (BNPP: c and d), and the of wood  $(P_W)$ , aboveground litter  $(P_{al})$ , residual  $(P_{resid})$ , root  $(P_r)$ , and belowground litter (Pbl) production to TNPP for mixed-grasslands of the Cuiabá Basin (left-panels) and Cerrado forests of the Pantanal (right-panels).  $P_W =$  wood production;  $P_{al} =$  aboveground litter production;  $P_{resid} =$  residual production (herbaceous production + production losses to herbivory and volatile organic C emission);  $P_r =$  root production;  $P_{bl} =$  belowground litter production.



Fig. 6. Conceptual carbon balance for the Cerrado forest in the Pantanal (left-panel) and the mixed grassland in the Cuiabá Basin (right-panel) based on a combination of stand increment and eddy covariance measurements for 2011-13 (Cuiabá Basin) and 2015-17 (Pantanal). Fluxes (MgC ha<sup>-1</sup> y<sup>-1</sup>) are displayed as arrows, with gains to the ecosystem as green arrows and losses as red arrows. Pools (boxes: MgC/ha) are based on data published by Pinto Jr. et al. (2020), Vourlitis et al. (2013, 2015, and 2019), and from unpublished data. Data are averages of those displayed in Table 1. The change in soil organic C (SOC) is estimated as the sum of the above and belowground litter

inputs minus heterotrophic respiration. The change in above- and below ground C pools (AGC and BGC, respectively) are estimated as NEE - SOC and are partitioned into AGC and BGC pools based on their fraction of the total plant C pool. Positive values of net ecosystem  $\rm CO_2$  exchange (NEE) depict net up take by the ecosystem.